Author’s response to reviews

**Title:** Do ultrathin strut bare-metal stents with passive coating improve efficacy in large coronary arteries? Insights from the randomized, multicenter BASKET-PROVE trials

**Authors:**

Kim Wadt Hansen (kim.wadt.hansen@regionh.dk)

Raban Jeger (Raban.Jeger@usb.ch)

Rikke Sørensen (Rikke.Soerensen@regionh.dk)

Christoph Kaiser (Christoph.Kaiser@usb.ch)

Matthias Pfisterer (matthias.pfisterer@bluewin.ch)

Tor Biering-Sørensen (tor.biering@gmail.com)

Louise Bjerking (louise_bjerking@hotmail.com)

Søren Galatius (galatius@dadlnet.dk)

**Version:** 3 **Date:** 10 Sep 2019

Author’s response to reviews:

1. Ethics approval

We have included a list of all local ethics committees including reference numbers in the Ethics approval and consent to participate section.

2. Clean manuscript

We have attached the revised, clean manuscript.